Author's response to reviews

Title: Media suicide-reports, Internet use and the occurrence of suicides between 1987 and 2005 in Japan

Authors:

Akihito Hagihara (hagihara@hsmp.med.kyushu-u.ac.jp)
Kimio Tarumi (tarumi_kimio@jpm1960.org)
Takeru Abe (takeru-a@hsmp.med.kyushu-u.ac.jp)

Version: 5  Date: 8 November 2007

Author's response to reviews:

As requested, we made corrections on formatting manuscript, except one below:
We did not have anything to mention in a section of Acknowledgements, so that we excluded the section.